Physics 221A Fall 2011 Homework 11 Due Monday, November 14, 2011

Reading Assignment:

Notes 19, Secs. 5–10, Notes 20 entire, Notes 21 entire.

This homework is due on Monday, November 14, since Friday, November 11 is officially a university holiday. There will be a discussion section as usual on Thursday, November 10. There will be no lecture or office hour on Friday, but I will hold an office hour on Monday, November 11, 1-2pm. The next homework will be due on Friday, November 18.

Please do Problems 19.2, 20.1, 21.1 and 21.2.